Specific Comments to Authors: I have no specific comments regarding the present erratum.

Reply: Thank you for your comment.

This is an erratum to an already published paper named “Establishment of a prediction model for prehospital return of spontaneous circulation in out-of-hospital patients with cardiac arrest” (Wang JJ, Zhou Q, Huang ZH, et al. Establishment of a prediction model for prehospital return of spontaneous circulation in out-of-hospital patients with cardiac arrest. World J Cardiol. 2023;15(10):508-517. doi:10.4330/wjc.v15.i10.508). We found errors in the affiliated institution of the authors section. We apologize for our unintentional mistake. Please note, these changes do not affect our results.

**Correction:** the affiliated institution of the authors

Jingjing Wang, Qiang Zhou, Zhenhua Huang, Yong Han, Chongzhen Qin, Zhongqing Chen, Xiaoyong Xiao, Zhe Deng

**Jingjing Wang, Zhenhua Huang, Yong Han, Zhongqing Chen, Xiaoyong Xiao, Zhe Deng,**
Department of Emergency Medicine, Shenzhen Second People’s Hospital / the First Affiliated Hospital of Shenzhen University Health Science Center, Shenzhen 518035, Guangdong, China.

**Qiang Zhou, Chongzhen Qin,** Shenzhen Emergency Medical Center, Shenzhen 518000, Guangdong, China.

Co-first authors: Jing-Jing Wang and Qiang Zhou